

September 1, 2014

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 12142-review.doc).

Title: Retinal Emboli

Authors: Michelle J. Kim, Donald S. Fong

Name of Journal: *World Journal of Ophthalmology*

ESPS Manuscript NO: 12142

The manuscript has been improved according to the suggestions of reviewers:

1. The manuscript has been extensively re-written to provide more analysis and critical discussion of the reviewed articles. This has led to the authors taking the stance that the mainstay of the management of retinal emboli should be medical therapy, with consideration for further workup and surgical management only in special circumstances.
2. The conclusion has been re-written and condensed to focus on just the main points of the review.
3. The number of references from the last 5 years has been increased from 3/32 (9%) to 14/43 (33%). Although it was advised that 70% of the references are from the last 5 years, this was not feasible in this case due to the stagnant nature of new papers published on this topic. The authors felt that to achieve the 70% target would introduce too many irrelevant points.
4. The entire manuscript has been carefully proofread for linguistic errors.
5. The postcode was added to the cover page.
6. Each author's contribution to the manuscript was added to the cover page.
7. Key words have been added.
8. Reference numbers have been changed to square brackets in superscript.

Thank you again for publishing our manuscript in the *World Journal of Ophthalmology*.

Sincerely,

Michelle J. Kim, MD
1011 Baldwin Park Blvd.
Baldwin Park, CA 91706
Phone: 626-851-6105
Fax: 626-564-6106
Email: jangmichellekim@gmail.com